Author’s response to reviews

Title: Balancing the struggle to live with dementia: A systematic meta-synthesis of coping

Authors:

Guro Bjørkløf (guro.bjorklof@aldringoghelse.no)
Anne-Sofie Helvik (anne-sofie.helvik@ntnu.no)
Tanja Ibsen (tanja.ibsen@aldringoghelse.no)
Elisabeth Telenius (elisabeth.w.telenius@aldringoghelse.no)
Ellen Grov (ellgro@oslomet.no)
Siren Eriksen (siren.eriksen@aldringoghelse.no)

Version: 1 Date: 12 Feb 2019

Author’s response to reviews:

Dear editor,

Thank you very much for the opportunity to submit our article. The corrections received at the 10th of February has now been thoroughly reviewed and the improvements made.

1. The method section and parts of the analysis section has been re-written
2. The email addresses for all authors are included on the title page.
3. The individual authors' contributions are described in the Authors’ Contributions section.
4. The figure legend is transferred after the references in a section called “figure legends”.
5. Introduction has been renamed to Background.

Yours sincerely,

Prof. Siren Eriksen